

Australian Government

Australian Institute of Health and Welfare



# Diabetes technology in Australia: scoping report

# Diabetes technology in Australia: scoping report

Australian Institute of Health and Welfare Canberra Cat. no. CVD 94 The Australian Institute of Health and Welfare is a major national agency whose purpose is to create authoritative and accessible information and statistics that inform decisions and improve the health and welfare of all Australians.

© Australian Institute of Health and Welfare 2021 (cc) EX

This product, excluding the AIHW logo, Commonwealth Coat of Arms and any material owned by a third party or protected by a trademark, has been released under a Creative Commons BY 3.0 (CC-BY 3.0) licence. Excluded material owned by third parties may include, for example, design and layout, images obtained under licence from third parties and signatures. We have made all reasonable efforts to identify and label material owned by third parties.

You may distribute, remix and build upon this work. However, you must attribute the AIHW as the copyright holder of the work in compliance with our attribution policy available at </www.aihw.gov.au/copyright/>. The full terms and conditions of this licence are available at </http://creativecommons.org/licenses/by/3.0/au/>.

A complete list of the Institute's publications is available from the Institute's website <www.aihw.gov.au>.

ISBN 978-1-76054-895-7 (Online) ISBN 978-1-76054-896-4 (PDF) DOI 10.25816/9qaa-9704

#### Suggested citation

Australian Institute of Health and Welfare 2021: Diabetes technology in Australia: scoping report. Cat. no. CVD 94. Canberra: AIHW.

#### Australian Institute of Health and Welfare

Board Chair Mrs Louise Markus Chief Executive Officer Mr Rob Heferen

Any enquiries relating to copyright or comments on this publication should be directed to: Australian Institute of Health and Welfare GPO Box 570 Canberra ACT 2601 Tel: (02) 6244 1000 Email: info@aihw.gov.au

Published by the Australian Institute of Health and Welfare

This publication is printed in accordance with ISO 14001 (Environmental Management Systems) and ISO 9001 (Quality Management Systems). The paper is sourced from sustainably managed certified forests.



Please note that there is the potential for minor revisions of data in this report. Please check the online version at <www.aihw.gov.au> for any amendments.

# Contents

Su	mmary	v
	Data gaps and future opportunities	v
	Approval, access and distribution pathways	vi
1	Background	1
	Structure of this report	2
2	What is diabetes?	3
	Treatment and management of diabetes	3
	Glucose monitoring	4
	Insulin therapy	5
	Diabetes-related complications	5
3	Diabetes technology	6
	Continuous glucose monitoring	7
	Outcomes for people with type 1 diabetes	7
	Outcomes for people with type 2 diabetes	8
	Flash glucose monitoring	8
	Outcomes for people with type 1 diabetes	9
	Outcomes for people with type 2 diabetes	9
	Continuous subcutaneous insulin infusions	10
	Non-automated insulin pumps	11
	Automated insulin delivery systems	12
	Information technology	12
	Outcomes for people with type 1 diabetes	12
	Outcomes for people with type 2 diabetes	13
4	Approval, access and distribution pathways for diabetes devices	16
	Approval of diabetes devices	17
	Health technology assessment	21
	Access and funding of diabetes devices	22
5	Monitoring diabetes technology use in Australia	24
	Why is it important to monitor diabetes technology use?	24
	Key areas to monitor	24
	Devices used in the management of diabetes	24
	Number of users of diabetes technology	24
	Characteristics of the users of diabetes technology	25

Experiences of the users of diabetes technologies	25
Short-term and long-term diabetes complications	25
Adverse events	25
Data sources available to monitor diabetes technology use	26
Administrative data sources	26
Registry data sources	29
Surveys	31
Other data sources	32
Data gaps	32
Experiences of users	33
Long-term diabetes-related complications	33
Scope and coverage of data sources	33
6 Future opportunities and recommendations	35
Baseline reporting	35
Filling data gaps	35
Data linkage	36
Conclusion	36
Acknowledgments	38
Abbreviations	39
Glossary	40
References	42
List of tables	46
List of figures	47

# Summary

Diabetes is a chronic condition in which the body cannot properly use its main energy source—the sugar glucose—resulting in high levels of glucose in the blood. It is caused either by the inability of the body to produce insulin (a hormone made by the pancreas to manage blood glucose levels) or by the body not being able to use insulin effectively, or both.

Management for people with diabetes includes healthy eating, regular physical activity and ongoing monitoring of glucose levels to minimise the risk of diabetes-related complications. All people with type 1 diabetes, and some people with type 2 diabetes, gestational diabetes and other forms of diabetes will also require insulin to manage their condition. Some people with diabetes may require other medications to maintain glucose levels within their target range.

Technology is transforming care for people with diabetes. It is increasingly being used for management of the condition. The technology available can be categorised in 3 broad groups: technology for monitoring glucose; technology for insulin delivery; and information technology for managing diabetes.

This report will assess and describe the current data gaps and currently available data sources for monitoring the use of technology (devices) in managing diabetes. The report will also briefly outline the approval, access and distribution pathways for diabetes technology in Australia.

### Data gaps and future opportunities

The data landscape has changed in recent years, with information about the uptake and characteristics of the users being captured in a range of administrative data sets and outcomes in clinical registries. Information about the number and characteristics of diabetes technology users, and adverse events associated with the use of technology, is being captured in some administrative data sets and clinical outcomes in clinical registries. However, there are gaps in information about the use of diabetes technology among the diabetes population and the impact of the technology on diabetes-related complications. Data gaps identified include the lack of nationally representative data sources, and the inability to comprehensively report across key monitoring areas including the trends in the uptake of devices for the management of diabetes; characteristics and experiences of diabetes technology users; and the impact on short-term and long-term diabetes-related complications.

Undertaking data improvement activities such as reporting using available data sources, modifying or developing data sources and linkage of data sources will require considerable resources. Future work will need to consider carefully the availability of data for reporting across key areas from existing data sources, balanced against the costs and benefits of undertaking data activities to improve the monitoring of the use of technology in managing diabetes.

### Approval, access and distribution pathways

The availability of and access to effective technology for Australians with diabetes has expanded in recent years. The approval, access and distribution pathways of new diabetes technology involve multiple stakeholders and processes. While there is a rigorous approval process for most diabetes management technology before it is available on the market in Australia, public information regarding the approval, access and distribution pathways for diabetes technology is limited.

# 1 Background

Diabetes that does not meet management goals can lead to the onset of diabetes-related complications including heart disease, stroke, kidney disease, retinopathy (loss of vision), neuropathy (nerve damage) and lower limb amputation. Maintaining glucose levels in range can prevent and reduce these complications.

Recently, the uptake of technology for managing diabetes to improve self-management as an adjunct to conventional therapy has increased. The availability of and access to these technologies has changed in recent times and a range of technology for monitoring glucose levels, such as continuous glucose monitoring (CGM), flash glucose monitoring (flash GM), and self-management tools such as mobile phone apps, is used. Other devices, such as insulin pumps, have been available for the last few decades in Australia as an alternative to the traditional method of injecting insulin with a pen or syringe.

The Australian Institute of Health and Welfare (AIHW) previously reported on the use of insulin pumps for people with type 1 diabetes in Australia (AIHW 2012). That report provided information regarding the number and characteristics of people using insulin pumps, and insights into their experiences. Since the *Insulin pump use in Australia* (2012) report, there has been an accelerated uptake of other technology for monitoring glucose levels and self-management tools, and changes in the access to these technologies.

There is a lack of comprehensive national data regarding the use of technology to manage diabetes and the impact of the technology on short-term and long-term diabetes-related complications, and the approval, access and distribution pathways for these technologies. The data landscape has changed in recent years, with some technology for delivering insulin and monitoring glucose being subsidised by the Australian Government, and information about usage and characteristics of users being captured in administrative data sets. With the increasing uptake of technology, clinical registries capturing information about diabetes care now also collect information on technology use.

Comprehensive information is needed to continue to build the evidence base about technology use in diabetes care, and its impact on the health of people with diabetes at the national level, and the formulation of policies and guidelines for diabetes care and access to technology.

This report describes the data sources currently available for monitoring the use of technology in the management and treatment of diabetes; identifies gaps; and provides recommendations for future work. The report also provides a brief outline of the approval, access and distribution pathways for diabetes technology.

### Structure of this report

This report has 6 chapters:

- Chapter 1 outlines the purpose and structure of the report.
- Chapter 2 describes diabetes and provides an introduction to its management and treatment.
- Chapter 3 outlines the technology available for diabetes management.
- Chapter 4 describes the approval, access and distribution pathways for diabetes technology.
- Chapter 5 discusses the importance of monitoring the use of technology for diabetes care, and describes the data sources currently available to monitor the use of technology.
- Chapter 6 discusses data gaps and provides recommendations for improving the monitoring of technology use in managing diabetes.

# 2 What is diabetes?

Diabetes is a chronic condition marked by high levels of glucose in the blood. It is caused either by the inability of the body to produce insulin (a hormone made by the pancreas to control blood glucose levels) or by the body not being able to use insulin effectively, or both.

The main types of diabetes are:

- type 1 diabetes—a lifelong auto-immune disease that can occur at any age
- type 2 diabetes—most common form of diabetes, often associated with lifestyle factors and generally of later onset. Risk is also associated with genetic and family-related risk factors
- gestational diabetes—glucose intolerance of varying severity, which develops or is first recognised during pregnancy
- other types of diabetes—typically related to certain conditions or syndromes which result in defects in insulin secretion, insulin action, or both.

### **Treatment and management of diabetes**

Management for people with diabetes includes healthy eating, regular physical activity and ongoing monitoring of glucose levels to minimise the risk of short-term and long-term diabetes-related complications. Some people with diabetes may require medications to maintain in-range glucose levels. Keeping blood glucose levels within an individual's target range requires a careful balance between glucose monitoring, diet, physical activity and medication.

All people with type 1 diabetes, and some people with type 2 diabetes, gestational diabetes and other forms of diabetes will also require insulin therapy to manage their condition.

#### Treatment of diabetes by type

#### Type 1 diabetes

A person with type 1 diabetes requires daily insulin replacement to survive, except in cases where a pancreatic transplant occurs.

#### Type 2 diabetes

Managed with changes to diet and exercise, oral hypoglycaemic (glucose-lowering) drugs such as sulfonylureas, non-insulin injectable glucose-lowering medications, insulin injections, or a combination of these methods.

#### **Gestational diabetes**

Often managed with changes to diet and exercise, while some cases require treatment with insulin and/or oral hypoglycaemic medications.

#### Other types of diabetes

For some people with other types of diabetes, adequate glycaemic management can be achieved through diet and exercise or the use of other medications. Some, however, may also require insulin to manage their glucose levels.

#### **Glucose monitoring**

Monitoring glucose levels, also known as self-monitoring, is an important part of diabetes management. It provides information about how food, physical activity, stress, hormones and medications are affecting glucose levels and helps people with diabetes to change their management if their glucose levels are not within their target range.

#### **Blood glucose monitoring**

Self-monitoring of blood glucose involves checking blood glucose levels. Recommendations for glucose management plans for people with diabetes vary according to the type of diabetes and the medications and/or insulin being used (RACGP 2020).

#### Monitoring of blood glucose by type of diabetes

#### Type 1 diabetes

Monitoring of blood glucose levels regularly (up to 6 times every day or as directed by a doctor) is recommended (Diabetes Australia 2020c).

#### Type 2 diabetes

Monitoring of blood glucose is recommended:

- for people on insulin and sulfonylureas
- for people not on insulin but having difficulty achieving glucose levels in their target range
- when monitoring hypoglycaemia or hyperglycaemia arising from another illness
- during pre-pregnancy and pregnancy
- when there is a clinical need for monitoring (for example, during changes in management or lifestyle)
- when glycosylated haemoglobin (HbA1c) estimates are unreliable.

It is not recommended that people with type 2 diabetes who are considered low risk and using oral glucose-lowering drugs (with the exception of sulfonylureas which carry a risk of inducing hypoglycaemia) undertake routine self-monitoring of blood glucose. The method and frequency of monitoring for type 2 diabetes depends on individual circumstances and therapeutic aims (RACGP 2020).

Checking blood glucose levels usually involves pricking the tip of the finger and placing a drop of blood onto a blood glucose strip which is inserted into a blood glucose monitor, which then displays the blood glucose level.

#### Continuous and flash glucose monitoring

In the last 10 years, monitors measuring interstitial glucose (fluid in the tissue) have been available in Australia. The 2 options are continuous glucose monitoring (CGM) and flash glucose monitoring (flash GM) devices which both use a sensor inserted just under the skin. These devices are increasingly used by people with (predominantly type 1) diabetes. (RACGP 2020).

#### Insulin therapy

Conventional insulin therapy involves multiple daily injections. Insulin is injected through the skin into the fatty tissue known as the subcutaneous layer, throughout the day.

Insulin can be injected in different ways: by insulin syringe, an insulin pen with a fine needle, or an insulin pump (Shah et al. 2016).

### **Diabetes-related complications**

The objective of diabetes management and treatment is to keep glucose levels within an individual's target range. When levels fluctuate beyond a recommended range, there is increased risk of short-term and long-term diabetes-related complications.

Short-term complications include diabetes ketoacidosis (a life-threatening condition due to lack of insulin, leading to a profound disturbance of the metabolism and a build-up of acids in the blood); hypoglycaemia (low blood sugar); hyperglycaemia (high blood sugar); increased susceptibility to infections; and reduced ability to heal.

Long-term diabetes-related complications may include heart disease, stroke, kidney disease, retinopathy, neuropathy and lower limb amputation.

## 3 Diabetes technology

Technology is transforming care for people with diabetes. It is increasingly being used for management of the disease. The technology can be broadly categorised into 3 main groups:

- 1. Technology for monitoring glucose—such as blood glucose monitoring, CGM and flash GM, which provide information about glucose patterns.
- 2. Technology for insulin delivery—such as insulin pens, syringes, continuous subcutaneous infusion of insulin (insulin pumps).
- 3. Information technology for management of diabetes—such as mobile phone apps, text messaging, wearable technology (for example, fitness trackers), web-based programs and clinic-based chronic condition care programs (RACGP 2020).

Some technologies have different components which have different functions (such as monitoring blood glucose levels, or medication delivery), and can fit more than 1 category.

The Australian Evidence-Based Clinical Guidelines for Diabetes provide recommendations for the use of medical technology to manage type 1 diabetes by health professionals and people living with diabetes, and guide shared decision making for treatment (Living Evidence for Diabetes Consortium 2020). In addition, the Australian Diabetes Society, the Australian Diabetes Educators Association, the Australasian Paediatric Endocrine Group and the Australasian Diabetes in Pregnancy Society Working Group have developed an Australian consensus statement for the implementation of technologies for people with type 1 diabetes. The Working Group recommended a shift from a focus on strict glycaemia criteria towards engagement and individualised management goals for commencing or continuing diabetes management technologies (Pease et al. 2021).

A comprehensive assessment of the evidence regarding the benefits and harms, the outcomes and key issues (such as equitable access, acceptability and feasibility) are presented in the Australian Evidence-Based Clinical Guidelines for Diabetes for the use of medical technology in managing type 1 diabetes (Living Evidence for Diabetes Consortium 2020). A range of studies investigate the impacts of the use of technology in managing diabetes on short- and long-term outcomes. A detailed analysis of the evidence regarding these impacts was not undertaken for this report, and information has been summarised from the published studies in Table 3.1.

The evidence regarding the impact of the use of technology on outcomes summarised in the sections below is drawn mainly from randomised controlled trials, cohort studies and metaanalysis or review of these studies. Randomised controlled trials are the gold standard for assessing the impact of an intervention, due to the lack of bias. However, randomised controlled trials are often relatively short in duration and might not mimic the treatment and management of diabetes in everyday life. Other sources such as administrative data sets and clinical registers present an opportunity to examine the effects of the technology on long-term diabetes-related complications and the impact of technology on experiences and outcomes.

Due to limited evidence regarding the use of technology in the management of gestational diabetes and other types of diabetes, this chapter focuses on type 1 and type 2 diabetes.

### Continuous glucose monitoring

CGM devices continuously monitor blood glucose levels, taking glucose measurements at regular intervals 24 hours a day. They collect information and provide real-time and summary glucose data. This real-time information allows people with diabetes to make timely decisions regarding management, whether by adjusting their insulin therapy or changing behaviour to achieve in-range glucose levels (Reddy et al. 2020).

The CGM system sensor is inserted under the skin to measure glucose in the fluid in the tissue. A transmitter is attached to the sensor and sends glucose readings through Bluetooth to a receiver (such as a standalone reader device, compatible insulin pump or smart phone apps) that allows people to view their glucose data (Figure 3.1). CGM devices have alarms that alert the user to out-of-range blood glucose levels, and can also be integrated with compatible insulin pumps. Data collected can be shared with health care professionals through secure Cloud-based platforms. CGM reduces the need to check blood glucose levels by pricking the tip of the finger and using a blood glucose monitor. However, finger prick checks are required at least twice daily for calibration (Diabetes Australia 2017).



The Australian Evidence-Based Clinical Guidelines for Diabetes recommends CGM instead of self-monitoring of blood glucose for adults with type 1 diabetes who are using multiple daily injections (Living Evidence for Diabetes Consortium 2020). CGM devices may also be used in the management of other diabetes such as type 2, especially for those at high risk of hypoglycaemia, with impaired awareness of hypoglycaemia and with high glucose variability (RACGP 2020).

#### Outcomes for people with type 1 diabetes

Randomised controlled trials in the United States and Germany have shown that adults with type 1 diabetes using multiple daily insulin injections and CGM with alerts see a decrease in HbA1c, increase in the time in target glucose range and decrease in severe hypoglycaemia when compared with self-monitoring blood glucose (Beck et al. 2017; Jendle & Heinemann 2018; Riddlesworth et al. 2017).

Since 2017, the Australian Government has provided access to fully subsidised CGM for eligible people. For more information see 'Chapter 4 Approval, access and distribution pathways for diabetes devices'. A recent study in Australia examined the impact of CGM at a large tertiary paediatric diabetes centre between May 2017 and December 2019 following the subsidisation of CGM devices for people under 21 years of age with type 1 diabetes.

The study found that people ceasing CGM use was high, with 28% and 33% of people ceasing use at 12 and 24 months after CGM commencement. People cited a dislike for wearing the device; technical and alarm issues; and site issues as reasons for ceasing use. There was no change in HbA1c levels before or after the use of CGM over the 24-month period. There was a reduction in the rate of severe hypoglycaemia over the 24 months but this was not statistically significant (Swaney et al. 2020).

The impact of CGM on long-term diabetes-related complications was not investigated by randomised controlled trials or cohort studies examining the impact of CGM in people with type 1 diabetes.

#### Outcomes for people with type 2 diabetes

The evidence from a narrative review of randomised controlled trials and observational trials also showed that CGM reduced HbA1c, bodyweight and caloric intake; promoted increased maintenance of personal eating plans; and increased physical activity compared with self-monitoring of blood glucose in adults with type 2 diabetes (Taylor et al. 2018). However, results should be interpreted with caution as the studies included in the review had small sample sizes and confounding from adjunctive counselling, education techniques and clinical involvement.

Relatively few studies have investigated the effects of CGM on outcomes in people with type 2 diabetes, and most are short in duration. Hence the use of CGM and associated impacts on short-term and long-term diabetes-related complications for adults with type 2 diabetes have not been investigated to a significant extent and are not supported by robust clinical evidence.

### Flash glucose monitoring

Flash GM is a sensor technology that measures glucose in the fluid in the tissue and collects information about the patterns in glucose levels. A small disc is positioned on the back of the upper arm, holding a sensor that is inserted under the skin into the subcutaneous tissue. The generation of flash GM in Australia at present requires the user to scan over the sensor (with either a mobile phone open to the compatible app), or a standalone reader device to give a glucose reading (Figure 3.2). Data are gathered in a smart phone app and can be shared with health professionals through a secure Cloud-based platform. Currently, these devices do not alert the user to either hypoglycaemia or hyperglycaemia without scanning (Diabetes Australia 2017).



### Outcomes for people with type 1 diabetes

A randomised controlled trial in Australia showed that flash GM did not result in greater improvements in glucose management than self-monitoring of blood glucose in youths with type 1 diabetes and high-risk glucose management. However, the use of flash GM resulted in higher glucose checking frequency and diabetes treatment satisfaction in youths with type 1 diabetes and high-risk glucose management (Boucher et al. 2020).

The use of flash GM devices in people with type 1 diabetes treated with multiple daily injections was associated with reductions in the time and frequency of hypoglycaemia compared with self-monitoring of blood glucose in another multi-centre randomised controlled trial that included 23 European diabetes centres (Oskarsson et al. 2018).

In Australia, the use of flash GM in people with type 1 diabetes with a recently confirmed episode of clinically significant hypoglycaemia did not reduce the incidence of severe hypoglycaemia when compared with people with type 1 diabetes who were self-monitoring their blood glucose levels (Davis et al. 2020).

International randomised controlled trials examining the impact of flash GM on outcomes were short in duration and have not investigated the effects of the technology on long term diabetes-related complications.

#### Outcomes for people with type 2 diabetes

Systematic review and meta-analysis of 2 randomised controlled trials have shown that flash GM use was associated with a non-significant reduction in HbA1c compared with self-monitoring of blood glucose in people with type 2 diabetes (Castellana et al. 2020).

In Australia, a randomised controlled trial (GP-OSMOTIC study) found that people with type 2 diabetes using flash GM had lower HbA1c levels than those with usual clinical care at 6 months after commencement of flash GM. There were no differences in HbA1c level in those using flash GM compared with those on usual care at 12 months, suggesting any effect in HbA1c reduction is short-term. However, glucose data were not available to participants in the study until their general practice visit, where the results were discussed

with a health professional (Furler et al. 2020). The same study also found that GP consultations increased in those using flash GM in the 6 months following its use (McMorrow et al. 2021). Another study in Australia found that there were no differences in the incidence of severe hypoglycaemia in people with insulin-treated type 2 diabetes with a recently confirmed episode of clinically significant hypoglycaemia and using flash GM compared with those self-monitoring their blood glucose levels (Davis et al. 2020).

Relatively few studies have investigated the effects of flash GM on outcomes in people with type 2 diabetes and none of these studies investigated its effect on long-term diabetes-related complications.

### **Continuous subcutaneous insulin infusions**

Continuous subcutaneous insulin infusions pump devices, more commonly known as insulin pumps, are computerised devices that deliver insulin subcutaneously through a plastic tube attached to the body. An insulin pump is a small battery-operated electronic device that holds a reservoir of insulin. It is programmed to deliver insulin into the body through thin tubing known as an infusion set, and is worn 24 hours a day outside the body. The infusion set has a fine needle or flexible Teflon cannula that is inserted just below the skin where it stays in place for 2 to 3 days (Figure 3.3). Whenever food is eaten the pump is programmed to deliver a surge of insulin into the body similar to the way the pancreas does in people without diabetes. Between meals a small and steady rate of insulin is delivered (Diabetes Australia 2020d).



In recent years, insulin pumps have been integrated with CGM. There are 2 broad categories of insulin pumps:

- non-automated insulin pumps with or without CGM (including low glucose suspend or predictive low glucose suspend functions that suspend insulin delivery)
- automated insulin delivery systems, also known as 'hybrid closed loop insulin pumps'.

Insulin pumps are being used increasingly to manage type 1 diabetes. However, factors such as high acquisition costs, ongoing costs for insulin pump consumables, and not wanting to be attached to a device influence the use of pumps in people with type 1 diabetes (Pyrlis et al. 2019).

The Australian Evidence-Based Clinical Guidelines for Diabetes recommends the use of an insulin pump or multiple daily injections for children and adolescents based on the preference of the person with type 1 diabetes and the carer. In adults with type 1 diabetes, it is suggested that insulin pump therapy is used rather than multiple daily injections (Living Evidence for Diabetes Consortium 2020).

#### Non-automated insulin pumps

Non-automated insulin pumps are worn 24 hours a day and the delivery of insulin through the pumps is controlled by the user. The basal dose is a continuous dose of insulin designed to keep blood glucose levels within the normal range between meals without inducing hypoglycaemia and the bolus dose is administered at meal times or when blood glucose levels are high, to prevent hyperglycaemia. Current pumps require the user to periodically review and modify settings for insulin basal rates based on their blood glucose levels and insulin requirements. Insulin boluses are adjusted manually by the user according to carbohydrates consumed at meal times or when blood glucose levels are high.

These devices are available with or without CGM. Various options include insulin pumps with low glucose suspend setting (the insulin pump suspends the delivery of insulin if low blood glucose is detected via the CGM) and predictive low glucose suspend setting (the insulin pump suspends the delivery of insulin if low blood glucose is detected or glucose trajectory predicts low blood glucose levels via the CGM).

#### Outcomes for people with type 1 diabetes

Meta-analysis and randomised controlled trials have shown that people with type 1 diabetes using non-automated insulin pumps have lower HbA1c levels, lower insulin requirements and improvements in glucose management compared with people using multiple daily injections (Benkhadra et al. 2017; Pickup et al. 2002). One cohort study showed that the non-automated insulin pump compared with multiple daily insulin injections, and was associated with lower risks of severe hypoglycaemia and diabetes ketoacidosis among young people with type 1 diabetes in Germany (Karges et al. 2017).

There is limited evidence on the effects of non-automated insulin pumps on long-term diabetes-related complications. Cohort studies have shown that non-automated insulin pump therapy was associated with lower cardiovascular mortality and lower rates of retinopathy and peripheral nerve abnormality in people with type 1 diabetes in Europe (Marchand et al. 2017; Steineck et al. 2015; Zabeen et al. 2016).

#### Outcomes for people with type 2 diabetes

The evidence surrounding the use of insulin pumps for people with type 2 diabetes is mixed. A meta-analysis of randomised controlled trials has shown that non-automated insulin pumps did not significantly improve HbA1c levels and the rate of hypoglycaemic episodes in comparison with multiple daily injections (Monami et al. 2009). Smaller multi-centre randomised controlled trials have shown that people with type 2 diabetes using non-automated insulin pumps had lower HbA1c and lower insulin requirements than people treated with multiple daily injections in Canada, Europe, Israel, South Africa, and the United States (Bode 2010; Reznik et al. 2014).

#### Automated insulin delivery systems

In recent years, insulin pumps have been integrated with CGM. These combined technologies are known as automated insulin delivery systems, or hybrid-closed loop systems. An automated insulin pump works like an artificial pancreas: glucose levels are monitored by CGM and a computerised algorithm automatically adjusts insulin delivery based on glucose readings. In most cases, automated insulin pumps deliver basal doses of insulin automatically based on glucose measurements from the CGM. Insulin boluses are adjusted manually by the user. Current studies are assessing the ability of new algorithms to deliver insulin boluses automatically. Fully automated insulin delivery systems are not available outside clinical trials.

The hybrid closed loop system, which automates all background or basal insulin but requires the user to adjust manually for bolus insulin when they consume carbohydrates, is the only device currently available for the management of diabetes.

Several do-it-yourself (DIY) diabetes technologies are also used by people to manage their condition. Due to the limited availability of automated insulin delivery systems, DIY technology such as open-source closed loop systems are being developed by people with diabetes, or parents of children with diabetes. These DIY technologies are not approved for use through the regulatory bodies such as the Therapeutic Goods Administration (TGA) in Australia (Diabetes Australia 2018).

#### Outcomes for people with type 1 diabetes

Randomised controlled trials have found that hybrid closed loop systems increased the time spent in target glucose range and reduced HbA1c levels and the risk of hypoglycaemia in people with type 1 diabetes, compared with non-automated insulin delivery systems, in France, the United Kingdom and the United States (Benhamou et al. 2019; Tauschmann et al. 2018). The impact of hybrid closed loop systems on long-term complications was not investigated in these short-duration trials.

### Information technology

Information technology such as mobile phone apps, text messaging, wearable technology (for example, smartwatches), web-based programs and clinical-based programs are increasingly being used in the management of diabetes.

These technologies include simple applications providing general information, reminders, documentation of clinical statistics, and data sharing with health care professionals and others (such as carers), to more complex analysis of blood glucose level patterns to derive insulin delivery recommendations and glucose predictions.

#### Outcomes for people with type 1 diabetes

A systematic review and meta-analysis of cohort studies has shown that mobile phone apps had variable effects on HbA1c, following glucose monitoring goals and hypoglycaemia in people with type 1 diabetes (Sun et al. 2019; Wu et al. 2019).

#### Outcomes for people with type 2 diabetes

A systematic review of cohort studies showed that mobile phone apps and web-based applications combined with standard diabetes care resulted in reductions in HbA1c in people with type 2 diabetes (Yoshida et al. 2018). Based on findings from a randomised controlled trial, a meta-analysis of randomised controlled trials and a narrative review, there is emerging evidence that information technology interventions are associated with reduced sedentary time through mobile and wearable technologies, increased physical activity (online self-tracking program) and improvements in diet in people with type 2 diabetes (Kooiman et al. 2018; Rollo et al. 2016; Stephenson et al. 2017). A systematic review and meta-analysis of cohort studies also found that mobile health interventions such as text messaging and applications reduced body weight and waist circumference, and improved glucose management in people with type 2 diabetes (Bacigalupo et al. 2013; Cai et al. 2020).

# Table 3.1: Summary of evidence on short-term complications related to diabetes technology used in the management of diabetes

Technology for monitoring blood glucose				
Type of technology	Description	Impact on short-term complications in people with type 1 diabetes	Impact on short-term complications in people with type 2 diabetes	
Continuous Glucose Monitoring (CGM)	Involves a small sensor being inserted under the skin to monitor glucose in the fluid in the tissue. 'Real-time' CGM continuously records and reports glucose levels, with devices using alarms to alert users to out of range glucose levels.	Decreases in HbA1c, improvements in glucose management and reductions in severe hypoglycaemia compared with self-monitoring blood glucose.	Reduced bodyweight and caloric intake; increases in following of a personal eating plan and increases in physical activity compared with self- monitoring of blood glucose.	
Flash Glucose Monitoring (flash GM)	Involves a small disc with a sensor inserted under the skin and worn on the arm. The sensor is scanned with a reader or mobile phone app to read interstitial glucose results.	Reduction or no effect on glucose management compared with self-monitoring of blood glucose in people with type 1 diabetes.	Reduction or no effect on HbA1c compared with self-monitoring of blood glucose.	
Technology for delivering in	sulin			
Non-automated insulin delivery systems (insulin pumps)	Non-automated insulin delivery systems are worn 24 hours a day. Basal insulin is programmed into the pump and delivered throughout the day, and bolus insulin is given by the user at meal times or to address high glucose levels.	Lower HbA1c, lower insulin requirements and improvements in glucose management compared with multiple daily insulin injections. Some evidence of lower cardiovascular mortality and lower rates of retinopathy and peripheral nerve abnormality.	Evidence surrounding the use of insulin pumps for type 2 diabetes is mixed. Some studies showed lower Hba1c levels and insulin requirements in comparison with multiple daily injections. Other studies did not show a reduction in HbA1c and hypoglycaemia.	
Automated insulin delivery systems	Some insulin pumps have interoperable CGM. The system works like an artificial pancreas, and glucose levels are monitored by the CGM and a computerised algorithm automatically adjusts basal insulin delivery based on glucose readings. Bolus insulin is given by the user at meal times. However, automated insulin delivery systems are not available for use outside of clinical trials. Hybrid closed loop systems, where the user manually adjusts for bolus insulin when they consume carbohydrates, is the only device currently available for the management of diabetes.	The hybrid closed loop system was associated with increased time spent in target glucose range, reduced HbA1c levels and the risk of hypoglycaemia compared with non-automated insulin delivery system.	No studies reporting on the use of Automated insulin delivery systems in people with type 2 diabetes.	

Table 3.1 (continued): Summary of evidence on short-term complications related to diabetes technology used in the management of diabetes

Information technology			
Mobile apps, web based applications and wearable technology	The function of the technology ranges from simple such as providing information and reminders to the more complex—analysis of glucose	Mobile phone apps had varying effects on HbA1c, following glucose monitoring goals and hypoglycaemia.	Mobile phone apps and web-based applications combined with standard diabetes care resulted in reductions in HbA1c.
patterns to derive insulin dose recommendations and glucose predictions.		Reduced sedentary time, increased physical activity and improved diet through mobile and wearable technologies.	

# 4 Approval, access and distribution pathways for diabetes devices

Diabetes technology is an emerging and growing area and access to effective technology for Australians with diabetes has expanded in recent years. While a rigorous process precedes the approval of most diabetes management technology before it is available on the market in Australia, there is limited public information regarding access pathways.

Key stakeholders including the Department of Health, the TGA, Diabetes Australia and the AIHW Diabetes Expert Advisory Group consulted to discuss approval and access pathways and to collate information regarding these processes.

Broadly, the approval, access and distribution of new technology involves the following processes:

- Where required, the manufacturer obtains a TGA conformity assessment certificate or provides market authorisation evidence from comparable overseas regulatory agencies to demonstrate appropriate conformity procedures have been applied to the device.
- The sponsor applies for the inclusion of new technology in the Australian Register of Therapeutic Goods (ARTG) and submits evidence to the TGA for pre-market approval and assessment (depending on the purpose and risk of the device). The sponsor may also be the manufacturer of the technology.
- The TGA approves new technology for inclusion in the ARTG if evidence shows that the technology is safe and performs as intended.
- Manufacturer or sponsor may also submit an application to undergo health technology assessment to seek a public or private subsidy under Australian Government Funding programs.
- The TGA conducts post-market surveillance of technology to ensure that the devices continue to meet safety standards and regulatory requirements.
- All sponsors of devices listed on the ARTG have a legal obligation to report adverse events to the TGA.
- Consumers access devices through Australian Government funding programs such as the National Diabetes Services Scheme (NDSS) or purchase devices directly from companies or through private health insurance.

A summary of the pathway is presented in Figure 4.1 and information regarding each process is provided in the sections below.



### Approval of diabetes devices

All medical devices, including insulin pumps, CGM and flash GM devices, must be approved by the TGA to be included in the ARTG before they can be supplied within, imported to, or exported from Australia.

This approval process consists of 2 components: a conformity assessment and an application to include the device in the ARTG. The person or company (sponsor) responsible for the importation, exportation and supply of the device in Australia is responsible for submitting an application to include the device in the ARTG. The sponsor may also be the manufacturer of the device. Before an application is submitted to include a medical device in the ARTG, a conformity assessment is required. Where a TGA conformity assessment is required, the manufacturer is responsible for submitting an application for conformity assessment certification (TGA 2020b).

#### What is a medical device?

A medical device is defined as an instrument, apparatus, appliance, software, implant, reagent, material or other article (whether used alone or in combination, and including the software necessary for its proper application) intended to be used for human beings for the purpose of 1 or more of the following:

- diagnosis, prevention, monitoring, prediction, prognosis, treatment or alleviation of disease
- diagnosis, monitoring, treatment, alleviation of or compensation for an injury or disability
- investigation, replacement or modification of the anatomy or of a physiological process or pathological process or state
- control or support of conception
- in vitro examination of a specimen derived from the human body for a specific medical purpose (TGA 2020c).

#### Pre-market assessment and approval

Pre-market approval consists of 2 components:

- 1. A conformity assessment—an independent assessment that a manufacturer has applied appropriate procedures to manufacture the device and these procedures comply with the regulatory requirements for safety, quality and performance.
- 2. An application to include the medical device in the ARTG.

The pre-market assessment and approval process for new medical devices differs according to the risk classification of the device. A classification system for devices based on the device's intended purpose, materials used in the device, degree of invasiveness in the human body and the duration of use has been adopted by the TGA. The manufacturer of the device is responsible for determining the risk classification by applying the TGA's risk classification rules (TGA 2020b). Table 4.1 presents examples of devices with different risk classifications.

Risk level	Classification(s)	Example
Low	Class I	Surgical retractors
		Continuous Glucose Monitoring transmitters <sup>(b)</sup>
Low to Medium	Class I – supplied sterile	Hypodermic needles
	Class I – with a measuring function Class IIa	Medicine cup with specific units of measurement
		Digital or infrared thermometers
Medium to High	Class IIb	Insulin pump <sup>(b)</sup>
		Continuous Glucose Monitoring receivers
High	Class III	Continuous Glucose Monitoring and Flash Glucose Monitoring sensors
High	Active implantable medical devices	Implantable defibrillator

#### Table 4.1: TGA medication device classification system<sup>(a)</sup>

(a) Information presented in the table was obtained from the TGA website and the ARTG (TGA 2020b).

(b) The device classification may be subject to changes as the TGA implements reforms including introducing new classification rules in the regulations (i.e. active medical devices, software based medical devices).

For all medical devices (other than Class I devices that do not have a measuring function and are not intended to be supplied sterile), manufacturers have to obtain a TGA conformity assessment certification or market authorisation evidence from comparable overseas regulatory agencies for their medical device before an application for the inclusion of the device in the ARTG can be submitted. The TGA conformity assessment process involves the manufacturer:

- determining the risk classification of the device
- applying the appropriate conformity assessment procedures to obtain conformity assessment evidence to demonstrate compliance with the Essential Principles of Safety and Performance (the manufacturer applies different conformity assessment procedures based on the risk classification of the device)
- providing this manufacturer's evidence to the TGA for the device to be included in ARTG (TGA 2020b).

Following this process, the sponsor submits an application for the inclusion of the medical device in the ARTG.

The approval process for diabetes management devices is complex and highly dependent on the risk classification of the device and the materials incorporated in the device. For example, CGM devices consist of a sensor kit, transmitter and receiver and each component has a different risk classification with different assessment and approval processes. CGM and flash GM sensors contain glucose oxidase from microbial origin, hence are Class III devices that require a conformity assessment before an application for their inclusion on the ARTG can be submitted.

The level of assessment required for a medical device before it is approved for inclusion in the ARTG by the TGA depends on the risk the device poses. Some lower-risk Class I medical devices can be included in the ARTG via the manufacturer's self-declaration without assessment, while high-risk medical devices must be assessed for quality, safety and performance by the TGA regardless of whether or not manufacturers have manufacturer's evidence from overseas comparable regulatory agencies. Once the medical device is approved for inclusion in the ARTG, it can be marketed for use in Australia (TGA 2020b).

#### Approvals of medical apps

In recent years, mobile apps to monitor blood glucose have become available for the management of diabetes. Depending on the function and intended purposes, some devices are regulated by the TGA. Mobile apps are considered medical devices when used for the diagnosis, prevention, monitoring, treatment or alleviation of disease.

The TGA has implemented reforms to the regulation of software-based medical devices, including software that functions as a medical device in its own right. The TGA has consulted with the technology sector to increase awareness of this requirement for medical apps (TGA 2020a). Changes to the regulation of software-based medical devices (including software as a medical device) took effect on 25 February 2021.

The TGA has published guidance documents that summarise the regulatory requirements for medical device software. The documents outline transition arrangements available for devices that may need to be reclassified or that qualify for an exemption or exclusion from the *Therapeutic Goods (Medical Devices) Regulations 2002*. The documents are available on the TGA website: https://www.tga.gov.au/regulation-software-based-medical-devices (TGA 2020a).

However, manufacturers may not be aware of changes and are encouraged to familiarise themselves with the legislative and regulatory requirements in Australia.

#### Post-market surveillance

After approval and inclusion in the ARTG, medical devices are monitored closely post-market to ensure that the devices continue to meet safety standards and regulatory requirements. Activities conducted for post-market surveillance of devices include:

- requiring sponsors (the person or company supplying the device in Australia) to report adverse events and other information within specific timeframes
- assessing and investigating reports of problems with medical devices submitted by consumers, health care professionals, the medical device industry and overseas signals
- checking evidence that medical devices are compliant (TGA 2020b).

The sponsor supplying the medical device is responsible for submitting annual reports from the manufacturer containing information such as the number of devices supplied, complaints received, and adverse event data to the TGA following the inclusion of a high-risk medical device (active implantable medical devices, implantable Class IIb medical devices, Class III medical devices, and Class 4 in vitro diagnostic medical devices) in the ARTG. Annual reports are submitted to the TGA in the first 3 years of the device being available on the market (TGA 2020b). Sponsors supplying diabetes devices such as Class III sensors for use with CGM and flash GM are required to provide annual reports to the TGA for post-market surveillance following the approval of these devices.

The TGA also monitors adverse events related to medical devices. In addition to reports from sponsors and manufacturers, adverse events reported by members of the public or health professionals are recorded in the Incident Reporting and Investigation Scheme database (IRIS). Information is transferred from the IRIS to the publicly available Database of Adverse Event Notifications (DAEN) after the TGA has verified the information reported in the IRIS. The DAEN provides general information on the medical device about the adverse event report. Information provided about the adverse event includes:

- an event description (a description of the adverse event)
- the reported event outcome (a description of the outcome reported to have been caused by the adverse event, for example injury)
- event type (general categories describing the type of adverse event as defined in the ISO/TS 19218-1:2011–Medical devices–Hierarchical coding structure for adverse events–Part 1: Event-type codes) (TGA 2018).

The TGA may also undertake a post-market review of a medical device at any time during inclusion in the ARTG. A post-market review can be initiated as a result of safety signals or a quality or performance issue from registry data; international regulatory notifications; publications in journals; or media articles. As part of a post-market review the sponsor will be required to provide evidence from the manufacturer to support the inclusion of the device in the ARTG. The evidence must demonstrate the continued compliance with the legislated requirements on the quality, safety, and performance of medical devices (TGA 2020d).

#### Health technology assessment

The manufacturer or sponsor of a diabetes technology may also submit an application for their technology to undergo a health technology assessment (HTA) to seek public or private subsidy under the Australian Government Funding programs. Applications for health technology assessment may be considered on a case-by-case basis for government funding by the HTA expert advisory committees. This process assesses the technology for comparative clinical and cost effectiveness compared with existing conventional interventions (Department of Health 2019).

The Australian Government has 3 HTA expert committees providing advice on whether health technologies should receive government funding:

- the Pharmaceutical Benefits Advisory Committee for pharmaceuticals to be funded under the Pharmaceutical Benefits Scheme and vaccines to be funded under the National Immunisation Program
- the Medical Services Advisory Committee (MSAC) for medical services involving new procedures or health technologies to be funded under the Medicare Benefits Schedule (MBS) or other programs
- the Prostheses List Advisory Committee for prostheses to be funded through private health insurance arrangements under the Prostheses List.

In recent years, technologies such as CGM and flash GM have been subsidised by the Australian Government for people with diabetes who meet the eligibility criteria. Diabetes technologies have been assessed by both the PBAC and MSAC for subsidisation in previous years. From November 2020, the MSAC is the designated health technology assessment body to consider diabetes-related products for subsidisation under the NDSS and the CGM Initiative (Department of Health 2020).

#### Access and funding of diabetes devices

There are a number of pathways for accessing and funding diabetes management technology. This section focuses on technologies used for monitoring blood glucose levels and insulin delivery, which are currently subsidised by the Australian Government for a sub-population of people with diabetes.

#### Insulin pumps

The current cost of an insulin pump is about \$10,000 (Paldus 2018). Eligible insulin pump users (including those with type 1 diabetes and gestational diabetes) can access subsidised insulin pump consumables under the NDSS. Subsidised insulin pump consumables cost an additional \$30 a month (Diabetes Australia 2017). The ways people with diabetes can access and fund insulin pumps are shown in Figure 4.2.

In Australia, people with diabetes can access insulin pump therapy through private health insurance. Currently, the Australian Government provides some funding for insulin pumps and children and adolescents with type 1 diabetes can access subsidised pumps through the Insulin Pump Program administered by the Juvenile Diabetes Research Foundation Australia (JDRF Australia). The program provides fully subsidised insulin pumps to low income families who have children aged under 18 with type 1 diabetes, and who do not have access to other forms of reimbursements such as private health insurance. In the 2018–19 Budget, the Australian Government provided additional funding to the Insulin Pump Program, which allowed for the number of pumps subsidised to increase from 68 to around 220 each year (JDRF 2020).

Adults with type 1 diabetes and people with type 2 diabetes and gestational diabetes who require insulin therapy do not have access to any subsidised programs for insulin pumps. However, insulin pump consumables (such as infusion sets and reservoirs) are subsidised under the NDSS for all people with type 1 diabetes. Insulin pump users who do not receive financial assistance pay the entire cost of the pump themselves or use a pump lent to them from the pump company. The cost of an insulin pump may be covered by private health insurance.

Most respondents to the 2011 Insulin Pump User Survey indicated that they received some kind of financial assistance to purchase insulin pumps, with 88% indicating this to be the case. Another 7% received no funding and paid for the cost of the insulin pump themselves or used a pump that had been lent to them (AIHW 2012). Of insulin pump users who received financial assistance, the majority (97%) received a rebate through private health insurance, 3% indicated they received an Australian Government subsidy and 2% a JDRF grant (AIHW 2012).

There is limited up-to-date information about the way people with diabetes access and fund the purchase of insulin pumps.



#### Continuous and flash glucose monitoring devices

The cost of CGM devices including the consumables is around \$4,000–5,000 per year (Diabetes Australia 2020a). The cost of flash GM products is around \$2,500 per year (Diabetes Australia 2020b). CGM and flash GM devices are unlikely to be covered by private health insurance. The cost of blood glucose meters may be covered by private health insurance in some cases (Diabetes Australia 2017).

Since 2017, the Australian Government has provided access to fully subsidised CGM devices and sensors for eligible people through the NDSS under the CGM Initiative. Flash GM devices were included under the CGM Initiative in 2020. The following groups are eligible for subsidised CGM and flash GM devices:

- children and young people with type 1 diabetes aged under 21
- people with type 1 diabetes aged 21 and over who have valid concessional status
- women with type 1 diabetes who are actively planning pregnancy, pregnant or immediately post-pregnancy
- children and young people aged under 21 with conditions very similar to type 1 diabetes who require insulin.

# 5 Monitoring diabetes technology use in Australia

# Why is it important to monitor diabetes technology use?

Diabetes technologies have been developed to improve glucose management and reduce the risk of diabetes-related complications. The use of technology for insulin delivery and glucose monitoring in the management of diabetes is increasing. However, there are gaps in information about the use of the technology among the diabetes population and the impact of the technology on long-term diabetes-related complications. Monitoring the use of technology for the management of diabetes is important for:

- assessing the impact of devices on the management of diabetes and short-term and long-term complications
- developing, implementing and evaluating policies for the subsidisation of new diabetes technology
- improving access to new diabetes technology
- providing recommendations and guidelines for the management of diabetes.

### Key areas to monitor

Comprehensive monitoring of the use of diabetes technologies requires specific data on the:

- type of device(s) used
- number of users
- characteristics of the users
- experiences of the users
- impact on short-term and long-term diabetes-related complications
- adverse events associated with the use of the technology.

#### Devices used in the management of diabetes

Data on the type of diabetes technology used is important for evaluating the characteristics and experiences of the users, the impact on short-term and long-term diabetes-related complications, and adverse events across the different categories of technology available.

#### Number of users of diabetes technology

Data on the number of users of diabetes technologies such as insulin pumps and CGM and flash GM devices over time is important for evaluating the uptake of new technology and the impact of new guidelines and introduction of subsidies on the use of these devices

The majority of people with diabetes will register with NDSS. Since 2017, the NDSS CGM initiative provides subsidised CGM to eligible cohorts of people with type 1 diabetes and other forms of diabetes that require insulin, or those who require controlled clinical management of their glucose levels. Flash GM products were included in the initiative from

2020 onwards. The NDSS could provide data on the number of people accessing subsidised products for insulin pumps, CGM and flash GM (only people with diabetes eligible for subsidised products) at a national level. Information about the number of people with diabetes currently accessing subsidised devices could be included in future reporting as data become available.

#### Characteristics of the users of diabetes technology

Demographic characteristics such as the age, sex, socioeconomic status and Indigenous status of users will allow for the identification of disparities in the access to devices for the management of diabetes.

#### Experiences of the users of diabetes technologies

Information on the factors, such as education and support, that influence decisions for starting, interrupting and ceasing using diabetes devices would provide a more accurate picture of the benefits and difficulties of using new diabetes management technology. It would also provide insights into the barriers to accessing new technology, such as the high costs associated with use and its impact on quality of life.

#### Short-term and long-term diabetes complications

While studies have shown that the use of technology for monitoring glucose levels and delivering insulin improved glucose management and decreased the frequency of hypoglycaemia and diabetes ketoacidosis, information about the impact of technology on reducing HbA1c levels over a long period, and subsequently long-term diabetes-related complications, is limited. There is also limited information about factors such as device acceptability, useability and adherence that affect the use of technologies, and the impact of technologies on behavioural changes (Taylor et al. 2018). It is important that these data are collected from device users over time to assess the long-term diabetes-related complications. These data will also help inform the development and evaluation of funding programs providing subsidised access to new technology, and clinical guidelines for the management of diabetes.

#### **Adverse events**

The TGA monitors adverse events related to medical devices and records of the adverse events are publicly available in the Database of Adverse Event Notifications (TGA 2018). While it is not mandatory for consumers or health professionals to report adverse events to the TGA, adverse event reporting is mandatory for sponsors or manufacturers of the medical devices.

The National Hospital Morbidity Database (NHMD) captures diagnostic information relevant only to the current episode of care—including conditions or disorders that affect the treatment received by the person during the episode. Diagnostic information relating to diabetes-related complications and mechanical complications due to the use of insulin pumps (for example, a broken insulin pump) are captured for an episode of care, and might provide some information about adverse events resulting from the use of pumps. However, mechanical issues with pumps or failures of monitoring sensors are more likely to occur outside the hospital setting and will not be captured in hospital data. Despite data being captured for some technology in some data sources, there are still limited national data on specific adverse events experienced by people using technology to manage their condition.

# Data sources available to monitor diabetes technology use

Data about technology use in, especially, people with type 1 diabetes are currently available from a number of sources in Australia.

Based on various methods of data collection, these data sources can be categorised into:

- administrative data
- survey-based data
- registry data sources
- other data sources.

Key stakeholders including Diabetes Australia and the AIHW Diabetes Expert Advisory Group consulted to discuss the data sources available for monitoring technology use. Information regarding the data sources presented in this chapter was based on discussions with stakeholders or publicly available information.

#### Administrative data sources

Administrative data sources contain information about the delivery of services, or a record of events, collected for historical or funding purposes. Data from administrative data sets such as the NDSS and Hospital Casemix Protocol data collection can be used for secondary purposes including monitoring the supply and usage of insulin pumps and CGM and flash GM devices. The NDSS and Hospital Casemix Protocol data collections are examined further in tables 5.1 and 5.2.

# Table 5.1: Assessing the National Diabetes Services Scheme for monitoring diabetes technology use

National Diabetes Services Scheme (NDSS)			
Type of data source		Administrative (national)	
Brief description		Established in 1987, the NDSS is an initiative of the Australian Government, administered with the assistance of Diabetes Australia. People with a diagnosis of diabetes by a health professional can be registered with the scheme. Once registered, they can access diabetes self-management information, services, and subsidised products—such as pens and needles to administer insulin, blood glucose strips, insulin pump consumables, and, if eligible, CGM and flash GM products.	
Geographic	coverage	Australia; state and territory	
Scope and coverage		Majority of people with diabetes will register with NDSS to access education material or subsidised products. However, registration with the NDSS is voluntary. All people with type 1 diabetes use insulin and are therefore most are likely to obtain subsidised products through the NDSS. Coverage of people with diabetes on the NDSS is thought to be relatively high, especially where diabetes-related products are required for management purposes.	
Limitations		Data related to the usage of CGM and flash GM products will only be available for the following groups eligible for subsidised products:	
		children and young people with type 1 diabetes aged under 21	
		people with type 1 diabetes and aged 21 and over who have valid concessional status	
		women with type 1 diabetes who are actively planning pregnancy, pregnant or immediately post-pregnancy	
		children and young people aged under 21 with conditions very similar to type 1 diabetes who require insulin.	
		NDSS Access Points may be limited in rural Australia and unavailable in remote communities, with other programs being available in these areas to assist with the purchase of diabetes-related products. As a result, the coverage of the NDSS may be lower in remote and very remote areas or across states and territories with large remote communities.	
		Aboriginal and Torres Strait Islander Australians can access diabetes-related products through other programs and this might result in lower registration for the NDSS, among Indigenous Australians.	
	Type of diabetes technology	Blood glucose monitoring consumables, insulin pump consumables, CGM and flash GM.	
	Number of users of diabetes technology	Trends data available.	
ı available	Characteristics of the users	Demographic information such as age, sex, date of birth, postcode and state of current residence, postcode and state of diagnosis, Indigenous status. Diagnosis information such as the type of diabetes and diagnosis date.	
Data	Experiences of the users	No data.	
	Short-term and long- term complications	No data.	
	Adverse events	No data.	

Table 5.2: Assessing the Hospital Casemix Protocol data collection for monitoring diabetes technology use

Hospital Casemix Protocol (HPC) data collection			
Type of	data source	Administrative (national).	
Brief description		The HCP data collection is a legislated data collection for all private health insurance funded admitted patient separations for which private health insurers have paid benefits. The data is collected by public and private hospitals, including day facilities, and then supplied to private health insurers, who in turn submit the data to the Department of Health.	
Geograp	hic coverage	Australia; state and territory	
Scope and coverage		The HCP data collection has records of all episodes of admitted patient care for which private health insurers have paid benefits. The collection has episodic, benefit and charge data for privately insured admitted patient episodes nationally from 1996–97.	
Limitations		Data are only available for privately insured patients starting a new pump or being fitted with an insulin pump in hospital. The data source also only provides information about the number of hospital separations for patients being fitted with an insulin pump.	
		admitted to hospital for the procedure.	
	Type of diabetes technology	Procedure information on patients being fitted with an insulin pump.	
D	Number of users of diabetes technology	Trends data available.	
Data available	Characteristics of the users	Patient demographic and clinical information.	
	Experiences of the users	No data.	
	Short-term and long- term complications	No data.	
	Adverse events	No data.	

#### **Registry data sources**

Registry data sources collect detailed information on persons with certain diseases or receiving particular treatment. Data from registries such as the Australasian Diabetes Data Network (ADDN) and Australian National Diabetes Audit (ANDA) can be used to report on the usage of insulin pumps and CGM and flash GM devices, as well as short-term and long-term complications. The ADDN and ANDA are examined further in tables 5.3 and 5.4.

Australasian Diabetes Data Network (ADDN)			
Type of	data source	Longitudinal register	
Brief description		ADDN is a research collaboration among Australia's largest diabetes centres to share clinical diabetes data. The ADDN registry holds longitudinal de-identified data to allow for the national surveillance of clinical outcomes.	
Geographic coverage		18 diabetes centres in New South Wales, Queensland, South Australia, Victoria, Western Australia and New Zealand.	
Scope and coverage		The ADDN registry holds clinical data for children and adolescents aged under 19 diagnosed with type 1 diabetes, type 2 diabetes, cystic fibrosis related diseases, monogenic diabetes, neonatal diabetes and other rarer forms of diabetes; and adults diagnosed with type 1 diabetes who have attended an ADDN centre for their diabetes clinical management.	
Limitations		While the ADDN registry provides data for paediatric and adult patients attending diabetes centres at the participating ADDN tertiary public hospitals, it is not representative of the entire diabetic population in Australia.	
	Type of diabetes technology	Information about CGM use such as usage time, start date and type of CGM device. Information about insulin pump use such as pump brand.	
	Number of users of diabetes technology	Trends data available.	
Data available	Characteristics of the users	Demographic information such as age, gender, date of birth, current postcode, postcode of diagnosis, ethnicity, and Indigenous status. Diagnosis information such as diabetes type and diagnosis date.	
	Experiences of the users	No data.	
	Short-term and long- term complications	CGM related measures such as mean glucose and time in target range of blood glucose in the last 2 weeks.	
		Diabetes ketoacidosis episodes, HbA1c measure at time of visit, severe hypoglycaemia, moderate hypoglycaemia	
	Adverse events	No data.	

 Table 5.3: Assessing the Australasian Diabetes Data Network for monitoring diabetes

 technology use

# Table 5.4: Assessing the Australian National Diabetes Audit – Australian Quality Clinical Audit for monitoring diabetes technology use

Australian National Diabetes Audit— Australian Quality Clinical Audit (ANDA—AQCA)			
Type of data source		Clinical quality registry	
Brief description		The ANDA is an annual diabetes audit that collects information from centres providing diabetes services across Australia. The audit provides participating diabetes services with individualised reports of their diabetes practice processes and patient outcome data in comparison with their peers.	
		The following two ANDA audits alternate each year:	
		Australian Quality Self-Management Audit focusing on self-management and diabetes distress and collects data related to diabetes education, self-care practices and quality of life.	
		ANDA—AQCA focusing on clinical indicators known to affect the care of the person with diabetes.	
		The addition of longitudinal linkage to the ANDA—AQCA in 2019 will allow for the analysis of trends in diabetes health outcomes and correlation to different treatment approaches.	
Geographic coverage		Participating diabetes centres across Australia. In 2019, the ANDA-AQCA collected data from 80 diabetes centres.	
Scope and coverage		The ANDA-AQCA clinical quality registry holds data on clinical indicators for the care of people with diabetes from participating primary to tertiary diabetes centres.	
Limitations		The ANDA-AQCA registry collects data from people attending participating diabetes centres and data are not representative of the diabetes population across Australia.	
		Most people referred to tertiary care diabetes services are referred by their general practitioners to receive specialist assessment and treatment. It is likely that people attending tertiary care diabetes services will be those with poorer glycaemic control, and with an increased number of comorbidities and diabetes-related complications.	
	Type of diabetes technology	Data for glucose monitoring (including the use of CGM).	
	Number of users of diabetes technology	Trends data available.	
ilable	Characteristics of the users	Demographic information such as age, gender and Indigenous status. Diabetes type and diagnosis date.	
ata ava	Experiences of the users	No data.	
Δ	Short-term and long-	Diabetic ketoacidosis and hypoglycaemic events.	
	term complications	Diabetes-related complications including eye disease, foot problems, kidney disease and vascular diseases.	
	Adverse events	No data.	

#### Surveys

Survey-based data sources collect health-related information through a population sample. Currently, information regarding the use of technology for the management of diabetes is not routinely collected in survey-based data sources such as the National Health Survey.

In 2011, the AIHW conducted a national survey of insulin pump users. The survey collected information relating to the experiences of those using pumps, including the benefits and problems with using them. The Insulin Pump User Survey is examined further in Table 5.5.

Insulin	Pumps User Survey	
Type of	data source	Survey
Brief description		The Insulin Pump User Survey was conducted by the AIHW in October–November 2011. The survey was developed to provide information about the factors that influenced people's choice to commence insulin pump therapy and the benefits and problems they experienced.
Geogra	ohic coverage	Australia
Scope and coverage		Participants were registrants on the NDSS, who were also eligible to purchase insulin pump consumables and had given consent to be contacted for research purposes. Nearly all participants on the scheme and who participated in this study have type 1 diabetes, though people with type 2 and gestational diabetes can be eligible in certain circumstances and may therefore have received and completed a survey.
		There were 9,618 people eligible to participate. The survey had a response rate of 59%.
Limitations		The Insulin Pump User Survey was a one-off national survey of insulin pump users and there have been substantial changes in the use of technology for managing diabetes.
	Type of diabetes technology	Information about insulin pump use.
	Number of users of diabetes technology	Number of users available. No trends data available.
Data available	Characteristics of the users	Demographic information such as age, sex and postcode of current residence. Age at diagnosis and age at start of insulin pump use.
	Experiences of the users	<ul> <li>Funding of insulin pumps, reasons for commencing insulin pump therapy and benefits of using an insulin pump.</li> <li>Problems with using the pump (hospitalisation or attendance at emergency department due to control issues).</li> <li>Breaks from insulin pump use and reasons for discontinuation.</li> </ul>
	Short-term and long-term complications	No data.
	Adverse events	No data.

Table 5.5: Assessing the Insulin Pumps User Survey for monitoring diabetes technology use

#### Other data sources

#### Data from private companies

Data regarding the purchase and sale of technology are available from private companies providing the technology. Demographic information and glucose management data for the use of CGM are also collected by companies supplying CGM devices in Australia. Further consultation with companies is required to determine the governance surrounding the use of the data and assess the quality of data for monitoring the use of technology to manage diabetes.

#### Juvenile Diabetes Research Foundation Australia

The JDRF Australia Insulin Pump Program provides subsidised insulin pumps to children aged under 18. The number of insulin pumps subsidised and the demographic characteristics of the children receiving them are likely to be captured by the JDRF Australia program (JDRF Australia 2020).

#### Fremantle Diabetes Study Phase II

The Fremantle Diabetes Study Phase II is a cohort study that recruited people with diabetes from the Fremantle hospital catchment area (postcode-defined population living in and around Fremantle in Western Australia) to examine clinically relevant aspects of diabetes including clinical management, metabolic control, complications and cost. Demographic information, diabetes-related information including the mode of insulin administration and clinical information including glucose management, HbA1c levels and complications were collected from consenting participants (Davis et al. 2013).

#### Database of Adverse Event Notifications – medical devices

The DAEN – medical devices collects information from reports of adverse events that the TGA receives in relation to medical devices. Information in the database comes from reports made to the TGA by patients, consumers, health professionals and manufacturers of the medical devices since July 2012.

Information about the total number of people using a device, the total number of adverse events, and the number of medical devices supplied in Australia are not captured on this database. Due to the limited information collected, the database cannot be used to determine the incidence of adverse events and the likelihood of a user experiencing that adverse event (TGA 2018).

### Data gaps

Data on the use of diabetes devices such as insulin pumps and CGM and flash GM devices are available from various data sources. Compared with CGM and flash GM, there is more detailed information about the use of insulin pumps. However, these data sources have their strengths and weaknesses and do not provide data to comprehensively monitor across key areas including the experiences of users and the impact of technology on short-term and long-term diabetes-related complications.

Information about the type of device used for insulin delivery and monitoring glucose levels is captured in most data sources. However, information about the use of information technology (such as mobile phone apps and wearable fitness trackers) for the management of diabetes is limited and rarely captured in the available data sources.

Data for the key monitoring areas are available from separate data sources and comprehensive national reporting of the use of diabetes technology from 1 data source is not possible. Furthermore, data sources collecting information on the use of diabetes technology do not capture all people with diabetes and it is difficult to monitor the use of diabetes technology across the population.

#### **Experiences of users**

Comprehensive data on the experiences of people with diabetes using technology for the management of their condition are not available from current administrative and registry data sources collecting data on the use of insulin pumps and CGM and flash GM devices.

A survey is probably necessary to collect this information. For example, the one-off Insulin Pump User Survey conducted in 2011 provided comprehensive information about the experiences of pump users.

#### Long-term diabetes-related complications

Clinical registry data sources such as the ADDN and ANDA collect data on glucose management and hypoglycaemia. Analysis of the data from these registries may allow associations to be drawn on the impact of the use of insulin pumps and CGM and flash GM devices on glucose management and subsequently reduction in hypoglycaemia. However, the scope and coverage of these clinical registries is restricted to people attending diabetes centres for treatment, and data on the use of these devices are collected mainly from people with type 1 diabetes.

Cohort studies have examined the association of the use of insulin pumps and long-term complications such as the progression of kidney disease, eye disease and vascular diseases. However, there is no national data source to monitor the impact of insulin pumps on long-term complications. Few studies have examined the effects of new technology such as CGM and flash GM devices on long-term diabetes-related complications.

There is currently no longitudinal national data source to monitor the outcomes of people with diabetes who use technology in their management of diabetes, especially the impact of new blood glucose monitoring technology.

Linkage of administrative data sources containing data on diabetes diagnosis, the use of technology and clinical outcomes to examine the impact of the use of technology on glycaemic control and subsequently long-term diabetes-related complications could be explored in the future. The potential use of data linkage to fill this gap in knowledge is discussed further in 'Chapter 6 Future opportunities and recommendations'.

#### Scope and coverage of data sources

No data sources presented in this report capture all people using technology as part of their diabetes management. This limits the ability to assess comprehensively the use of diabetes technology, and the experiences and outcomes of people using technology to manage their condition. A data set that captures all people with diabetes would also allow for the comparison of the experiences and outcomes of people using newer technology to manage their diabetes with those who manage their diabetes through conventional therapy.

The majority of people with diabetes will register with NDSS. The NDSS CGM initiative provides subsidised products to eligible cohorts of people with type 1 diabetes and other forms of diabetes that require insulin, or those who require controlled clinical management of their glucose levels—and will capture eligible people accessing subsidised CGM and Flash GM devices from the NDSS. However, there will be a proportion of people, such as those with type 2 diabetes, who are not eligible for subsidised products or not registered with the NDSS, and subsequently not captured in the administrative data set.

Other registry data sources such as the ADDN and ANDA collect data for a sub-population of people who have attended a diabetes care centre for treatment, but contain extensive data on the usage of insulin pumps and CGM and flash GM devices, as well as some data on short-term complications.

# 6 Future opportunities and recommendations

Based on the data gaps, data sources and information related to the use of diabetes technologies outlined in this report, the following opportunities for analysis and future work have been identified:

- Baseline reporting could be carried out, involving analysis of the currently available data sources outlined in this report, to look at the use of technology in the management of diabetes at the national level.
- Enhancing the existing data, including experiences of users, adverse events and longterm outcomes associated with the use of technology in the management of diabetes.
- Data linkage to bring together data to report comprehensively on the use of technology and longer-term outcomes of people with diabetes.

### **Baseline reporting**

The next steps in monitoring the use of technology to manage diabetes could be to undertake analysis to report on baseline estimates of the users of diabetes technology.

Analysis of data sources such as the NDSS could provide data on the number accessing subsidised products for insulin pumps, CGM and flash GM (only people eligible for subsidised products) at a national level; and the characteristics of the users of these devices. This would provide a national snapshot of the data available and could be used to facilitate further work in this area.

## Filling data gaps

Another next step could be engaging with key stakeholders to modify existing data sources or develop new data sources to capture information about the characteristics and experiences of users; complications; and adverse events. Modification or development of data sources should also consider expanding the information captured about the type of technology used in the management of diabetes to include information about the use of technology such as mobile phone apps and wearable technology. This data development work would require an assessment of information available in existing data sources (such as the adverse event data collected in the DAEN by the TGA) and additional information required for comprehensive reporting on the use of technology in the management of diabetes, before undertaking the work to modify or develop data sources.

Clinical registers, such as the ADDN, already capture most data regarding the use of technology to manage diabetes and short-term diabetes-related complications, to monitor the use of technology. These data sources could be modified to enable a more comprehensive capture of information, such as adverse events and experiences of the users.

In 2011, the AIHW conducted an Insulin Pump Use Survey. Diabetes Australia, which maintains the list of registrants in the NDSS, distributed the survey to registrants using insulin pumps on behalf of the AIHW (AIHW 2012). A similar survey examining the use of different technology could be designed and distributed to all registrants with diabetes to examine their experiences with using technology to monitor their glucose and deliver insulin, and conventional therapy. This would allow for comprehensive reporting of the experiences of

users of diabetes technology including the factors that influence choice, and the benefits and difficulties of using technology for the management of their condition. Data from this survey could be extrapolated to the broader diabetes population.

Further investigations into the ability to access data from pharmaceutical and technology companies could be undertaken. Future work could examine the possibility of accessing detailed data regarding the purchase of technology for delivering insulin and glucose monitoring from these companies.

### Data linkage

There is no data source that can provide complete information on the use of diabetes technology, and the experiences and outcomes of the users of diabetes technology. Data linkage to bring together information from 2 or more data sources to determine associations between the use of diabetes technology for medicine delivery and monitoring of glucose levels and outcomes could be explored and considered.

The linkage of data sources such as the administrative NDSS data set, which captures most people with diabetes who are using technology to manage it, with emergency and hospital data could allow for the examination of the impact of the use of technology on severe hypoglycaemic or ketoacidosis hospitalisations and progression of diabetes-related complications. However, severe hypoglycaemia can be managed by family members/carers or via ambulance call out without presentation to a hospital, and information might not be comprehensively captured in hospital data sets. Nationally linked longitudinal data would also allow insights into the pathways and individual impacts to be assessed. As new technologies such as CGM and flash GM were subsidised through the NDSS CGM initiative only in 2017 and 2020 respectively, future work will also need to assess the quality and completeness of data on the usage of these diabetes devices for linkage.

### Conclusion

Diabetes increases the risk of diabetes-related complications such as heart disease, stroke, kidney disease, retinopathy, heart failure and limb amputation. Maintaining optimal glucose management can prevent and reduce the risk of these complications. In addition to conventional self-monitoring using blood glucose monitors and insulin delivery through injections or insulin pumps, a range of newer technologies is available to manage diabetes. Many information technologies such as mobile phone apps and web-based applications are also available.

In recent years, the availability of and access to technology to manage diabetes has changed and there is emerging evidence that technologies can prevent or reduce progression to diabetes-related complications.

The approval, access and distribution pathway of new technology involves multiple stakeholders and processes. While there are clear pathways for the approval of new technology in Australia through the TGA, this varies based on the intended purpose, materials used in the device, and degree of invasiveness of the device used in the management and treatment of diabetes. Information technology such as mobile apps and web-based applications are a particular challenge as manufacturers and companies might not be aware that these applications may require pre-market approval for distribution and use in Australia.

Access to new technology for monitoring glycaemia has changed in recent years with the subsidisation of CGM and flash GM devices by the Australian Government. New devices may be considered by the health technology assessment bodies on a case by case basis for public or private subsidy. In November 2020, the MSAC was designated the health technology assessment body to consider diabetes-related products for subsidisation under the NDSS and the CGM Initiative.

There are also limited national data about the use of diabetes technology in Australia. The data landscape has changed in recent years, with information about the uptake and characteristics of the users being captured in a range of administrative data sets, and outcomes in clinical registries. However, data on the characteristics, experiences and outcomes of users of diabetes devices are available in separate data sources and comprehensive national reporting from a single combined data source is not possible.

Undertaking reporting using currently available data sources such as the NDSS can provide baseline data on the number of people accessing subsidised products for insulin pumps, CGM and flash GM devices. This would provide a national snapshot of the users of the common technologies in diabetes care and could facilitate further work in this area. Given that currently no data source can provide complete information on the use of technology in Australia, opportunities for data linkage or modifying existing data sources could be explored.

While there are opportunities for improving the data for monitoring the use of technology for the management of diabetes, considerable resources are likely to be required to improve existing data sources or develop new ones. The development and availability of new technology, the uptake of new technology to manage diabetes in the clinical setting and the capture of information across administrative data sets and clinical registries are also changing over time. Comprehensive data for some key areas such as the characteristics of the users of technology, and short-term complications, are already available from existing data sources. Future work will need to consider the factors affecting the use of technology over time and data already available for reporting across the key areas, and explore the costs and benefits of undertaking further data improvement activities to report on the use of technology for managing diabetes.

# Acknowledgments

This scoping report was written by Pearl Ng from the Cardiovascular, Diabetes and Kidney Unit at the AIHW. Chloe Martin from the AIHW provided assistance with the literature review for the scoping report.

Valuable input, review and advice were provided by James Pearce, Miriam Lum On, Richard Juckes and Justin Harvey of the AIHW and individual staff members at the Therapeutic Goods Administration.

This scoping report was prepared under the guidance of the AIHW Diabetes Expert Advisory Group, whose members are Jonathan Shaw (chair), Maria Craig, Wendy Davis, Mark Harris, Greg Johnson, Renza Scibilia, Glynis Ross and Sophia Zoungas.

The Australian Government Department of Health funded this scoping report and the authors acknowledge the valuable comments from individual staff members at the department.

The AIHW would like to acknowledge the National Diabetes Services Scheme (NDSS) for approval to use selected images throughout this publication. The NDSS is an initiative of the Australian Government administered by Diabetes Australia. The AIHW would like to acknowledge Diabetes Victoria for approval to use the image of the insulin pump in this publication.

# Abbreviations

ADDN	Australasian Diabetes Data Network
AIHW	Australian Institute of Health and Welfare
ANDA	Australian National Diabetes Audit
ARTG	Australian Register of Therapeutic Goods
AQCA	Australian Quality Clinical Audit
CGM	Continuous glucose monitoring
DAEN	Database of Adverse Event Notifications
DIY	Do it yourself
Flash GM	Flash glucose monitoring
HbA1c	Glycosylated haemoglobin
HCP	Hospital Casemix Protocol
HTA	Health technology assessment
IRIS	Incident Reporting and Investigation Scheme
JDRF	Juvenile Diabetes Research Foundation
MSAC	Medical Services Advisory Committee
NDSS	National Diabetes Services Scheme
NHMD	National Hospital Morbidity Database
RACGP	Royal Australian College of General Practitioners
TGA	Therapeutic Goods Administration

# Glossary

**Basal rate (dose):** A continuous low level delivery of insulin, required to keep blood glucose levels within the normal range as the liver releases glucose into the blood.

**Blood glucose:** The main sugar that the body makes from the food in the diet. Glucose is carried through the bloodstream to provide energy to all cells in the body. Cells cannot use glucose without the help of insulin.

**Bolus dose:** A delivery of insulin that provides the boost of insulin needed to stop the rise in blood glucose levels that occurs after meals.

**Diabetes:** A chronic condition in which the body cannot properly use its main energy source: the sugar glucose. This is due to either the pancreas not producing enough of the hormone insulin, or the body being unable to effectively use the insulin produced. Insulin helps glucose enter the body's cells from the bloodstream and then be processed by them. Diabetes is marked by an abnormal build-up of glucose in the blood, and can have serious short- and long-term effects on many of the body's systems, especially the blood vessels and nerves.

**Gestational diabetes:** A form of diabetes that is first diagnosed during pregnancy (gestation). It may disappear after pregnancy but signals a high risk of diabetes occurring later on in life.

**Incidence:** Refers to the number of new cases of an illness, disease, or event occurring during a given period.

**Insulin:** A hormone produced by the pancreas. Its main action is to enable body cells to absorb glucose from the blood and use it for energy.

Insulin pump: A device worn close to the body that delivers insulin subcutaneously.

**HbA1c (glycosylated haemoglobin):** Can be used to assess the average blood glucose over the preceding 6–8 weeks and is considered the gold standard for assessing glycaemic control. Targets for HbA1c in people with diabetes should be individualised, but a general target of less than or equal to 7.0% is recommended for people with type 2 diabetes.

**Heart attack:** Life-threatening emergency that occurs when a vessel supplying blood to the heart muscle is suddenly blocked completely by a blood clot. The medical term commonly used for a heart attack is acute myocardial infarction.

**Heart failure:** A term that describes when the heart functions less effectively in pumping blood around the body. It can result from a wide variety of diseases and conditions that can impair or overload the heart, such as heart attack, or other conditions that damage the heart muscle directly.

**Hypoglycaemia (low blood glucose)**: A common and dangerous condition for many people with type 1 diabetes. It can be caused by eating less than usual, undertaking more exercise than normal or if too much insulin or sulfonylurea medication is administered.

**Hyperglycaemia (high blood glucose)**: Occurs when the body has too much food or glucose, or too little insulin. It can be caused by a blockage in insulin pump tubing, missing an insulin dose, eating more than usual, stress or less exercise than usual.

**Ketoacidosis:** when the body cannot turn glucose into energy (due to insufficient insulin) it uses its own fats for energy. When these fats are broken down they release acids known as ketones, which can accumulate in dangerous levels in the blood and urine, causing diabetes ketoacidosis. This condition is potentially life threatening if not treated.

**Low suspend glucose:** A setting within the insulin pump system to suspend basal insulin delivery in response to detected low glucose levels.

**Predictive low glucose suspends:** A setting within the insulin pump system to suspend basal insulin delivery in response to predicted low glucose levels.

**Other types of diabetes:** Other types of diabetes are relatively uncommon, and are most typically related to certain conditions or syndromes that result in defects in insulin secretion, insulin action, or both. For some people with other types of diabetes, adequate glycaemic control can be achieved through diet and exercise or use of diabetes non-insulin medications. Some, however, may also require insulin to manage their blood glucose.

**National Diabetes Service Scheme (NDSS)**: Started in 1987 and is an initiative of the Australian Government, administered by Diabetes Australia. The NDSS delivers diabetes-related products at subsidised prices and provides information and support services to people with diabetes.

**Neuropathy:** A complication of diabetes that results in damage to the nerves.

**Retinopathy:** A complication of diabetes that causes damage to the capillaries of the retina in the eye.

**Procedure:** A clinical intervention that is surgical in nature, carries a procedural risk, carries an anaesthetic risk, requires specialised training and/or requires special facilities or equipment available only in an acute-care setting.

**Statistical significance:** Variation or difference in observed values or rates may reflect only a random variation or difference. Statistical significance assesses whether differences in values of rates are statistically significant—that is, that they are likely to not be due to chance alone.

**Stroke:** An event that occurs when an artery supplying blood to the brain suddenly becomes blocked or bleeds. A stroke often causes paralysis of parts of the body normally controlled by that area of the brain, or speech problems and other symptoms.

**Type 1 diabetes:** Marked by the inability to produce insulin. People with type 1 diabetes need insulin replacement for survival. Most cases are caused by an autoimmune condition that destroys the pancreatic cells which produce insulin.

**Type 2 diabetes:** The most common form of **diabetes**, occurring mostly in people aged 40 and over, and marked by reduced or less effective insulin.

## References

AIHW (Australian Institute of Health and Welfare) 2012. Insulin pump use in Australia. Diabetes series no.18. Cat. no. CVD 58. Canberra: AIHW.

Bacigalupo R, Cudd P, Littlewood C, Bissell P, Hawley MS & Buckley Woods H 2013. Interventions employing mobile technology for overweight and obesity: an early systematic review of randomized controlled trials. Obesity Reviews. 14(4):279–91.

Beck RW, Riddlesworth T, Ruedy K, Ahmann A, Bergenstal R, Haller S et al. 2017. Effect of continuous glucose monitoring on glycemic control in adults with type 1 diabetes using insulin injections: the DIAMOND randomized clinical trial. Journal of the American Medical Association 317(4):371–78.

Benhamou P-Y, Franc S, Reznik Y, Thivolet C, Schaepelynck P, Renard E et al. 2019. Closed-loop insulin delivery in adults with type 1 diabetes in real-life conditions: a 12-week multicentre, open-label randomised controlled crossover trial. The Lancet Digital Health 1(1):e17–25.

Benkhadra K, Alahdab F, Tamhane SU, McCoy RG, Prokop LJ & Murad MH 2017. Continuous subcutaneous insulin infusion versus multiple daily injections in individuals with type 1 diabetes: a systematic review and meta-analysis. Endocrine 55(1):77–84.

Bode BW 2010. Insulin pump use in type 2 diabetes. Diabetes Technology and Therapeutics 12 (Supplement 1):S17–21.

Boucher SE, Gray AR, Wiltshire EJ, de Bock MI, Galland BC, Tomlinson PA et al. 2020. Effect of 6 months of flash glucose monitoring in youth with type 1 diabetes and high-risk control: a randomized controlled trial. Diabetes care 43(10):2388–95.

Cai X, Qiu S, Luo D, Wang L, Lu Y & Li M 2020. Mobile application interventions and weight loss in type 2 diabetes: a meta-analysis. Obesity (Silver Spring) 28(3):502–9.

Castellana M, Parisi C, Di Molfetta S, Di Gioia L, Natalicchio A, Perrini S et al. 2020. Efficacy and safety of flash glucose monitoring in patients with type 1 and type 2 diabetes: a systematic review and meta-analysis. BMJ Open Diabetes Research and Care 8(1): e001092.

Department of Health 2019. About Health Technology Assessment. Canberra: Department of Health. Viewed 5 December 2020.

https://www1.health.gov.au/internet/hta/publishing.nsf/Content/about-1

Department of Health 2020. National Diabetes Services Scheme (NDSS). Canberra: Department of Health. Viewed 4 December 2020. https://www.health.gov.au/initiatives-and-programs/national-diabetes-services-scheme-ndss

Davis TME, Bruce DG & Davis W 2013. Cohort Profile: The Fremantle Diabetes Study. International Journal of Epidemiology 42(2):412–21.

Davis TME, Dwyer P, England M, Fegan PG & Davis W 2020. Efficacy of intermittently scanned continuous glucose monitoring in the prevention of recurrent severe hypoglycaemia. Diabetes Technology and Therapeutics 22(5):367–73.

Diabetes Australia 2017. Glucose self-monitoring in adults with type 1 or type 2 diabetes positions statement. Canberra: Diabetes Australia. Viewed 26 May 2021. https://www.diabetesaustralia.com.au/research-advocacy/position-statements/ Diabetes Australia 2018. People with type 1 diabetes and Do It Yourself (DIY) technology solutions—position statement. Canberra: Diabetes Australia. Viewed 25 May 2021. https://www.diabetesaustralia.com.au/research-advocacy/position-statements/

Diabetes Australia 2020a. Continuous glucose monitoring—fact sheet. Canberra: Diabetes Australia. Viewed 20 November 2020. https://www.ndss.com.au/aboutdiabetes/resources/find-a-resource/continuous-glucose-monitoring-fact-sheet/

Diabetes Australia 2020b. Flash glucose monitoring—fact sheet. Canberra: Diabetes Australia. Viewed 20 November 2020. https://www.ndss.com.au/about-diabetes/resources/find-a-resource/flash-glucose-monitoring-fact-sheet/

Diabetes Australia 2020c. Managing type 1 diabetes. Canberra: Diabetes Australia. Viewed 18 November 2020. https://www.diabetesaustralia.com.au/living-with-diabetes/managing-your-diabetes/managing-type-1/

Diabetes Australia 2020d. Insulin pumps. Canberra: Diabetes Australia. Viewed 26 May 2021. https://www.diabetesaustralia.com.au/living-with-diabetes/managing-your-diabetes/diabetes-technology/insulin-pumps/

Diabetes Victoria 2013. Understanding insulin pumps. Melbourne: Diabetes Victoria. Viewed 11 August 2021. https://www.diabetesaustralia.com.au/wp-content/uploads/Understanding-insulin-pumps.pdf

Furler JF, O'Neal D, Speight J, Blackberry I, Manski-Nankervis JA, Thuraisingam S et al. 2020. Use of professional-mode flash glucose monitoring, at 3-month intervals, in adults with type 2 diabetes in general practice (GP-OSMOTIC): a pragmatic, open-label, 12-month, randomised controlled trial. The Lancet Diabetes and Endocrinology 8(10):17–26.

JDRF Australia (Juvenile Diabetes Research Foundation Australia) 2020. Insulin Pump Program. JDRF. Viewed 19 November 2020. https://jdrf.org.au/living-with-t1-diabetes/insulinpump-program/

Jendle J & Heinemann L 2018. Real-time continuous glucose monitoring usage in pilots with diabetes: an option to improve safety. Diabetes Technology and Therapeutics 20(7):453–4.

Karges B, Schwandt A, Heidtmann B, Kordonouri O, Binder E, Schierloh U et al. 2017. Association of insulin pump therapy vs insulin injection therapy with severe hypoglycemia, ketoacidosis, and glycemic control among children, adolescents, and young adults with type 1 diabetes. Journal of the American Medical Association 318(14):1358–66.

Kooiman TJM, de Groot M, Hoogenberg K, Krijnen WP, van der Schans CP & Kooy A 2018. Self-tracking of physical activity in people with type 2 diabetes: a randomized controlled trial. Computers Informatics Nursing 36(7):340–9.

Living Evidence for Diabetes Consortium 2020. Australian evidence-based clinical guidelines for diabetes. Living Evidence for Diabetes Consortium. Viewed 26 November 2020. https://app.magicapp.org/#/guideline/4496

Marchand L, Kawasaki-Ogita Y, Place J, Fayolle C, Lauton D, Boulet F et al. 2017. Long-term effects of continuous subcutaneous insulin infusion on glucose control and microvascular complications in patients with Type 1 diabetes. Journal of Diabetes Science and Technology 11(5):924–9.

Monami M, Lamanna C, Marchionni N & Mannucci E 2009. Continuous subcutaneous insulin infusion versus multiple daily insulin injections in type 2 diabetes: a meta-analysis. Experimental and Clinical Endocrinology and Diabetes 117 (Supplement 1):220–2.

McMorrow R, Thuraisingam S, Furler J, & Manski-Nankervis J, 2021. Professional flash glucose monitoring and health service utilisation in type 2 diabetes: a secondary analysis of the GP-OSMOTIC study. Primary Care Diabetes 15(1):178–83.

Oskarsson P, Antuna R, Geelhoed-Duijvestijn P, Kröger J, Weitgasser R & Bolinder J 2018. Impact of flash glucose monitoring on hypoglycaemia in adults with type 1 diabetes managed with multiple daily injection therapy: a pre-specified subgroup analysis of the IMPACT randomised controlled trial. Diabetologia 61(3):539–50.

Paldus B, Lee MH & O'Neal DN 2018. Insulin pumps in general practice. Sydney: NPS Medicinewise.

Park C & Le QA 2018. The effectiveness of continuous glucose monitoring in patients with type 2 diabetes: a systematic review of literature and meta-analysis. Diabetes Technology Therapeutics 20(9):613–21.

Pease AJ, Andrikopoulos S, Abraham MB, Craig ME, Fenton B, Overland J et al. 2021. Utilisation, access and recommendations regarding technologies for people living with type 1 diabetes: consensus statement of the ADS/ADEA/APEG/ADIPS Working Group. Medical Journal of Australia doi:10.5694/mja2.51118.

Pickup J, Mattock M & Kerry S 2002. Glycaemic control with continuous subcutaneous insulin infusion compared with intensive insulin injections in patients with type 1 diabetes: meta-analysis of randomised controlled trials. British Medical Journal 324(7339):705. doi:10.1136/bmj.324.7339.705.

Pyrlis F, Brown F & Ekinci E 2019. Recent advances in management of type 2 diabetes. East Melbourne: The Royal Australian College of General Practitioners.

RACGP (Royal Australian College of General Practitioners) 2020. Management of type 2 diabetes: a handbook for general practice. East Melbourne: The Royal Australian College of General Practitioners.

Reddy N, Verma N and Dungan K 2020. Monitoring technologies—continuous glucose monitoring, mobile technology, biomarkers of glycemic control. Endotext, MD Text. Viewed 15 January 2021. https://www.ncbi.nlm.nih.gov/sites/books/NBK279046/

Reznik Y, Cohen O, Aronson R, Conget I, Runzis S, Castaneda J et al. 2014. Insulin pump treatment compared with multiple daily injections for treatment of type 2 diabetes (OpT2mise): a randomised open-label controlled trial. The Lancet 384(9950):1265-72.

Riddlesworth T, Price D, Cohen N & Beck RW 2017. Hypoglycemic event frequency and the effect of continuous glucose monitoring in adults with type 1 diabetes using multiple daily insulin injections. Diabetes therapy: research, treatment and education of diabetes and related disorders 8(4):947–51.

Rollo ME, Aguiar EJ, Williams RL, Wynne K, Kriss M, Callister R et al. 2016. eHealth technologies to support nutrition and physical activity behaviors in diabetes self-management. Diabetes, metabolic syndrome and obesity: targets and therapy 9:381–90.

Shah RB, Patel M, Maahs DM & Shah VN 2016. Insulin delivery methods: past, present and future. International Journal of Pharmaceutical Investigation 6(1):1–9.

Steineck I, Cederholm J, Eliasson B, Rawshani A, Eeg-Olofsson K, Svensson A-M et al. 2015. Insulin pump therapy, multiple daily injections, and cardiovascular mortality in 18,168 people with type 1 diabetes: observational study. British Medical Journal 350:h3234.

Stephenson A, McDonough SM, Murphy MH, Nugent CD & Mair JL 2017. Using computer, mobile and wearable technology enhanced interventions to reduce sedentary behaviour: a systematic review and meta-analysis. International Journal of Behavioral Nutrition and Physical Activity 14:105.

Sun C, Malcolm JC, Wong B, Shorr R & Doyle M-A 2019. Improving glycemic control in adults and children with type 1 diabetes with the use of smartphone-based mobile applications: a systematic review. Canadian Journal of Diabetes 43(1):51–8.e3.

Swaney EEK, McCombe J, Coggan B, Donath S, O'Connell MA & Cameron FJ 2020. Has subsidized continuous glucose monitoring improved outcomes in pediatric diabetes? Pediatric Diabetes 21(7):1292–1300.

Tauschmann M, Thabit H, Bally L, Allen JM, Hartnell S, Wilinska ME et al. 2018. Closed-loop insulin delivery in suboptimally controlled type 1 diabetes: a multicentre, 12-week randomised trial. The Lancet 392(10155):1321–29.

Taylor PJ, Thompson CH & Brinkworth GD 2018. Effectiveness and acceptability of continuous glucose monitoring for type 2 diabetes management: A narrative review. Journal of Diabetes Investigation 9(4):713–25.

TGA (Therapeutic Goods Administration) 2018. Database of Adverse Event Notifications Canberra: TGA. Viewed 26 November 2020. https://www.tga.gov.au/database-adverse-event-notifications-daen

TGA 2020a. Regulation of software as a medical device. Canberra: TGA. Viewed 25 November 2020. https://www.tga.gov.au/regulation-software-medical-device

TGA 2020b. Overview of medical devices and IVD regulation. Canberra: TGA. Viewed 25 November 2020. https://www.tga.gov.au/sme-assist/medical-devices-regulation-introduction

TGA 2020c. Update to medical device definitions and requirements for system or procedure packs. Canberra: TGA. Viewed 17 December 2020. https://www.tga.gov.au/update-medical-device-definitions-and-requirements-system-or-procedure-packs

TGA 2020d. Post-market reviews. Canberra: TGA. Viewed 25 May 2020. https://www.tga.gov.au/post-market-reviews

Wu X, Guo X & Zhang Z 2019. The efficacy of mobile phone apps for lifestyle modification in diabetes: systematic review and meta-analysis. Journal of Medical Internet Research mHealth and uHealth 7(1):e12297.

Yoshida Y, Boren SA, Soares J, Popescu M, Nielson SD & Simoes EJ 2018. Effect of health information technologies on glycemic control among patients with type 2 diabetes. Current Diabetes Report 18(12):130.

Zabeen B, Craig M, Virk S, Pryke A, Chan A, Cho YH et al. 2016. Insulin pump therapy Is associated with lower rates of retinopathy and peripheral nerve abnormality. PloS One 11(4):e0153033.

## List of tables

Table 3.1: Summary of evidence on short-term complications related to diabetes technology used in the management of diabetes	. 14
Table 4.1: TGA medication device classification system	. 19
Table 5.1: Assessing the National Diabetes Services Scheme for monitoring diabetes           technology use	. 27
Table 5.2: Assessing the Hospital Casemix Protocol data collection for monitoring diabetes           technology use	. 28
Table 5.3: Assessing the Australasian Diabetes Data Network for monitoring diabetes           technology use	. 29
Table 5.4: Assessing the Australian National Diabetes Audit - Australian Quality Clinical         Audit for monitoring diabetes technology use	. 30
Table 5.5: Assessing the Insulin Pumps User Survey for monitoring diabetes technology use .	. 31

# List of figures

Figure 3.1: Diagram of continuous glucose monitoring	. 7
Figure 3.2: Diagram of flash glucose monitoring	. 9
Figure 3.3: Diagram of insulin pump (without CGM)	10
Figure 4.1: Summary of the approval, access and distribution pathways for new technology used in the management of diabetes	17
Figure 4.2: Summary of the pathways for accessing and funding insulin pumps	23



The availability and access of technology for managing diabetes has changed. This scoping report describes the current data gaps and currently available data sources for monitoring the use of technology for the management of diabetes. The report also briefly outlines the approval, access and distribution pathways for diabetes technology in Australia.

aihw.gov.au



Stronger evidence, better decisions, improved health and welfare